Author’s response to reviews

Title: Descending Necrotizing Mediastinitis Caused by Streptococcus Constellatus in an Immunocompetent Patient: Case Report and Review of the Literature

Authors:

rui-hai ye (yehoo2003@163.com)

jun chao yang (yangjunchaozj@163.com)

hui hua hong (honghuiha@sina.com)

yu ping mao (2242723113@QQ.com)

yuan hong zhu (zhuyuan0487@126.com)

yu cao (caoyumail@163.com)

zhen wang (wangzhen610@sina.cn)

Version: 5 Date: 12 Dec 2019

Author’s response to reviews:

changes were made, but CT imagine of the mediastinal involvement still absent and the diagnosis have to be based on this imaging

Answer: 1. Contrast-enhanced CT showed a clear mediastinal abscesses. Figure 1E. Chest Enhanced Computed tomography of the mediastinum showing significantly low density for mediastinal abscesses with calcification.

2. Figure 2A. EBUS-TBNA of the mediastinum showing hypodense loculated fluid(long arrow) and mediastinal lymph node calcification(short arrows).

3. Figure 2E. EBUS-TBNA of the mediastinum revealing a foam-like purulent secretion of about 5ml(arrow).